Author's response to reviews

Title: Bilaterally Primary Cementless Total Hip Arthroplasty in Patients With Ankylosing Spondylitis

Authors:

Wanchun Wang (wangwc2013@126.com)
Guoliang Huang (huanggl2014@126.com)
Tianlong Huang (sarsevilwind@gmail.com)
Ren Wu (wurenraul@hotmail.com)

Version: 5 Date: 25 September 2014

Author's response to reviews:

Dear Sir or Madam:
Thank you for your attention to my article.

Comments:
"Thank you for your review. I have a minimal comment:
Has your article some relation with this other?. It is not mentioned in your references although the topic is quite similar.

Int Orthop. 2014 Jul 17.
Cementless bilateral synchronous total hip arthroplasty in ankylosing spondylitis with hip ankylosis.

Answer: Thank you. I read the article published in Int Orthop and mentioned it in our references.